Author’s response to reviews

Title: Prognostic value of hematogenous dissemination and biological profile of the tumor in early breast cancer patients. A prospective observational study

Authors:

Montserrat Solá (msola.germanstrias@gencat.net)
Mireia Margelí (33198mmv@comb.es)
Eva Castellá (ecastella.germanstrias@gencat.net)
Juan Francisco Julián (421127jii@gmail.com)
Miquel Rull (miquelrull@gmail.com)
Josep M Gubern (jgubern@csdm.cat)
Antonio Mariscal (mariscal.germanstrias@gencat.cat)
Agustí Barnadas (ABarnadasM@santpau.cat)
Manel Fraile (mfraile.germanstrias@gencat.cat)

Version: 4 Date: 6 June 2011

Author’s response to reviews: see over
Dear Tonilynn Manibo,

Together with this letter we resubmit the final version of our manuscript

Prognostic value of hematogenous dissemination and biological profile of the tumor in early breast cancer patients. A prospective observational study (MS:1215098591482621) to be reconsidered for publication in BMC Cancer.

We have changed the author list. Abstract has been reformatted to the correct structure. We have indicated all authors’ contributions and their approval to the final manuscript. Also, we have made minor revisions.

We hope that in its present form our manuscript will be acceptable for publication.

Montserrat Solá